Author’s response to reviews

Title: Systematic review on the needle and suture types for uterine compression sutures: A literature review.

Authors:
Shinya Matsuzaki (zacky@gyne.med.osaka-u.ac.jp)
Mariko Jitsumori (banri822j@yahoo.co.jp)
Takeya Hara (tttake.0303@gmail.com)
Satoko Matsuzaki (zacky05s@gmail.com)
Satoshi Nakagawa (s.nakagawa@gyne.med.osaka-u.ac.jp)
Tatsuya Miyake (tatsuyam0716@gmail.com)
Tsuyoshi Takiuchi (takkitakkitakki3@gmail.com)
Aiko Kakigano (ahayapyon1020@yahoo.co.jp)
Eiji Kobayashi (ekobayashi@gyne.med.osaka-u.ac.jp)
Takuji Tomimatsu (tomimatsu@gyne.med.osaka-u.ac.jp)
Tadashi Kimura (tadashi@gyne.med.osaka-u.ac.jp)

Version: 1 Date: 06 Apr 2019

Author’s response to reviews:

April 6, 2019, 2019
Hayley Henderson
Editor-in-Chief
BMC Surgery

Dear Editor:
Thank you for reviewing our manuscript titled “Systematic review on the needle and suture types for uterine compression sutures: A literature review.” The editor’s comments were very helpful, and we have revised our manuscript accordingly. Revised sentences were highlighted in blue font in the revised manuscript.

We would like to take this opportunity to express our sincere gratitude to the editor who identified areas of the manuscript that needed corrections or modifications. We have revised the
whole manuscript to improve the presentation. The manuscript has been carefully reviewed by an experienced editor whose first language is English and who specializes in editing papers written by scientists whose native language is not English. We would also like to thank you for allowing us the opportunity to resubmit a revised copy of the manuscript.

We hope that the revised manuscript will be suitable for publication as an Original Article in BMC Surgery. We look forward to hearing from you at your earliest convenience.

Sincerely yours,
Shinya Matsuzaki, MD, PhD
Department of Obstetrics and Gynecology
Osaka University Graduate School of Medicine
2-2 Yamadaoka, Suita, Osaka 565-0871, Japan
Tel No: +81-6-6879-3355; Fax: +81-6-6879-3359
E-mail: zacky@gyne.med.osaka-u.ac.jp